Author’s response to reviews

Title: The dating of thrombus organization in cases of pulmonary embolism: an autopsy study

Authors:

Celestino Sardu (drsarducele@gmail.com)
Gelsomina Mansueto (gelsomin.mansueto@unina.it)
Dario Costa (dario.costa@unicampania.it)
Emanuele Capasso (emanuele.capasso@unina.it)
Federica Varavallo (celestino.sardu@virgilio.it)
Giuseppina Brunitto (celestino.sardu@unicampania.it)
Rosanna Caserta (sarducelestino@libero.it)
Salvatore Esposito (drcelesardu@gmail.com)
Massimo Niola (massimo.niola@unina.it)
Celestino Sardu (drcelesardu@gmail.com)
Raffaele Marfella (raffaele.marfella@unicampania.it)
Claudio Napoli (claudio.napoli@unicampania.it)
Mariano Paternoster (mariano.paternoster@unina.it)

Version: 2 Date: 13 May 2019

Author’s response to reviews:

1. We note that the current submission contains potentially identifiable information. Currently, the age and gender information, in combination with other identifiers, may compromise patient/participant anonymity. Please amend your manuscript to address this issue, by removing this information.

We Thank the reviewer for his/her comments. As You suggested, we removed the current age informations from the text.
2. Please also clarify in the “Ethics approval and consent to participate” section of the Declarations whether Written consent was obtained from the next-of-kin of your patient to be able to use tissue samples in further research.

We thank the reviewer for his/her comments. As You suggested we reported this lost information in the “Ethics approval and consent to participate” section of the Declarations.

3. Additionally, we note that you state samples were routinely collected at the University of Naples Federico II. Please clarify whether the samples were collected from a hospital associated with the University, or whether samples were obtained from the university itself.

We thank the reviewer for his/her comments. As You suggested, we clarified that the samples were obtained from the Federico II university itself.

4. At this stage, we ask that you submit a clean version of your manuscript and do not include track changes or highlighting.

We thank the reviewer for his/her comments. As You suggested, we re-submitted a clean version of our manuscript, that dooe not include track changes or highlighting.